Author's response to reviews

Title: Short term micronutrient-antioxidant supplementation has no impact on gastric atrophy in Zambian adults: retrospective analysis of a randomised controlled trial

Authors:

Violet Kayamba (viojole@yahoo.com)
Mumba Chomba (mumbaschomba@gmail.com)
Paul Kelly (m.p.kelly@qmul.uk.ac)

Version: 2 Date: 24 July 2013

Author's response to reviews: see over
Dear Tonilynn Manibo,

Thank you for the advice on the editorial requirements.

1. I have included the Trial Registration Number at the end of the abstract submission system and in the main manuscript.
2. I have corrected Paul Kelly’s email address.
3. I have included in the text, a clear statement on the consent obtained. Both the participants and the University of Zambia ethics committee gave consent for these samples to be collected and analysed.

Yours Faithfully,

Violet Kayamba.